Author’s response to reviews

Title: Evaluating audio-visual falls prevention messages with community-dwelling older people using a World Café forum approach.

Authors:
Lex D. de Jong (lexddejong@gmail.com)
Jacqueline Francis-Coad (jacqui.francis-coad@nd.edu.au)
Chris Wortham (chris.wortham@nd.edu.au)
Terry Haines (terrence.haines@monash.edu)
Dawn Skelton (dawn.skelton@gcu.ac.uk)
Tammy Weselman (tammy.weselman@curtin.edu.au)
Anne-Marie Hill (anne-marie.hill@curtin.edu.au)

Version: 2 Date: 31 Oct 2019

Author’s response to reviews:

1. Please ensure the email address of the corresponding author is the same in the manuscript as in the submission system. Currently, they differ.
   - Corresponding author for submission and corresponding author for reprints both indicated on title page of the manuscript. The email address of the corresponding author for submission is now the same as in the submission system.

2. As a member of the editorial board (Dr. Anne-Marie Hill) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.
   - We have changed the Competing Interests section of the Declarations as suggested.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Please ensure all figures, tables and additional files are cited in the text, as they may be removed by the production department. Please delete any additional files not intended for publication.
   - All actions taken as requested.